Author’s response to reviews

Title: Antibacterial, antidiarrhoeal and cytotoxic activities of four fractions of Caesalpinia bonducella (L.) Roxb leaves

Authors:

Muhammad Mutassim Billah (faysol_billah@yahoo.com)
Md. Rafikul Islam (rafik_ruph@yahoo.com)
Mst. Hajera Khatun (hkhatun1@gmail.com)
S.M. Anisul Islam (anisuliiuc@gmail.com)
Md. Akbar Ali Mia (akbar4188@yahoo.com)
Dr. Ekramul Islam (ekrams74@yahoo.com.au)

Version: 2 Date: 3 September 2012

Author’s response to reviews: see over
Caesalpinia bonducella L. (Family: Fabaceae) is an important medicinal plant which is widely distributed in the tropical and subtropical regions of Asia and Caribbean (Chakrabarti S, Kinoshita T). The plant is known as Nata in Bengali and fever nut or nicker nut in English. The different parts of the plant have extensive uses in folk medicines for the treatment of a variety of diseases. Many of its traditional uses have been scientifically established. But there is no any report on antidiarrhoeal, antibacterial and cytotoxic activities of leaves of C. bonducella. So we have investigated those medicinal effects of methanol, ethyl acetate, chloroform and pet ether fractions of leaves and found better activities. Besides, we have concluded all the reported pharmacological works of the plant in our manuscript. Hence, we hope our finding is an important contribution in the research arena and expect your positive evaluation of our manuscript for your reputed journal.